Author's response to reviews

Title: Evaluation of alternate outreach models for cataract services in rural Nepal

Authors:

   Ram P Kandel (rpkandel@wlink.com.np)
   Sowmya R Rajshekaran (sowmyarajasekaran@gmail.com)
   Maria Gautam (kmmeh@wlink.com.np)
   Ken Bassett (bassett@chspr.ubc.ca)

Version: 4 Date: 11 March 2010

Author's response to reviews:

March 11, 2010

Editor-in-Chief
BioMedical Central Ophthalmology

Dear Sir/Madame

re: Evaluation of alternate outreach models for cataract services in rural Nepal

Thank you for your patience and kind support for our work. Please consider the changes to the format of the manuscript submitted for publication by BioMedical Central Ophthalmology.

We appreciate the opportunity to change the format of our manuscript.

Please find the point-by-point description of the changes made on page 2.

Yours truly,

Ram Prasad Kandel (Author for correspondence)
Program Director
Seva Nepal
Nepal Eye Hospital Compound, Tripureshwor
Kathmandu Nepal
Phone Number: 014218241
Fax: 014260558
E-mail: rpkandel@wlink.com.np

1. Used the preferred style in the listing of the authors.
2. Full affiliation of the authors written.
3. Removed unnecessary text’s in the title page.
4. Renamed Introduction as Background.
5. Renamed Conclusions and recommendations as Conclusions.
6. Inserted tables section directly after the references.
7. Removed the blank page from the manuscript.
8. Provided full citation details for reference 10.